Author’s response to reviews

Title: A novel compound heterozygous mutation of the SMARCAL1 gene leading to mild Schimke immune-osseous dysplasia: A case report

Authors:

Shuaimei Liu (719305071@qq.com)
Mingchao Zhang (zmchj99@163.com)
Mengxia Ni (nimengxia@126.com)
Peiran Zhu (980346785@qq.com)
Xinyi Xia (xixynju@163.com)

Version: 1 Date: 22 Jul 2016

Author’s response to reviews:

BPED-D-16-00371

A novel compound heterozygous mutation of SMARCAL1 gene leading to mild Schimke immune-osseous dysplasia: A case report

Shuaimei Liu; Mingchao Zhang; Mengxia Ni; Peiran Zhu; Xinyi Xia

BMC Pediatrics

Dear Dr Xia,

Thank you for submitting your manuscript, "A novel compound heterozygous mutation of SMARCAL1 gene leading to mild Schimke immune-osseous dysplasia: A case report" (BPED-D-16-00371), to BMC Pediatrics.

Before it can be sent out for review, please carry out the corrections, below:
(1) Manuscript Body: Please note that Case Presentation require the following sections:

- Background
- Case Presentation
- Conclusions

Please ensure that these sections are present and clearly labelled as described above. Please do check the instructions for authors on the journal website to ensure that your manuscript follows the correct structure for this journal and article type, and to ensure that you are aware of additional recommendations for formatting that will facilitate handling of your manuscript.


Please submit your revised manuscript by accessing the following site:

http://bped.edmgr.com/

If you have forgotten your username or password please use the "Send Login Details" link to get your login information. For security reasons, your password will be reset.

Please note, if your manuscript is accepted you will not be able to make any changes to the authors, or order of authors, of your manuscript once the editor has accepted your manuscript for publication. If you wish to make any changes to authorship before you resubmit your revisions, please reply to this email and ask for a 'Request for change in authorship' form which should be completed by all authors (including those to be removed) and returned to this email address. Please ensure that any changes in authorship fulfil the criteria for authorship as outlined in
BioMed Central’s editorial policies (http://www.biomedcentral.com/about/editorialpolicies#authorship).

Once you have completed and returned the form, your request will be considered and you will be advised whether the requested changes will be allowed.

By resubmitting your manuscript you confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.

Please be aware that we may investigate, or ask your institute to investigate, any unauthorised attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.

We look forward to receiving your revised manuscript before 28 Jul 2016.

Best wishes,

Joselito Acosta

Journal Editorial Office

BioMed Central

editorial@biomedcentral.com

www.biomedcentral.com

BMC series - open, inclusive and trusted.
Editorial Policies

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Pediatrics must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (http://www.biomedcentral.com/bmcpediatr).

Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

Declarations

- Ethics approval and consent to participate

- Consent to publish

- Availability of data and materials

- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information